

Observation of a Synchronicity between Shallow and Deep Seismic Activities during the Foreshock Crisis Preceding the Iquique Megathrust Earthquake

Michel Bouchon ()*1, Stéphane Guillot ()1, David Marsan ()1, Anne Socquet ()1, Jorge Jara ()2, François Renard

¹ISTerre, Université Grenoble Alpes, Université Savoie Mont Blanc, CNRS, IRD, Grenoble, France, ²German Research Center for Geosciences (GFZ), Potsdam, Germany, ³Njord Centre, Departments of Geosciences and Physics, University of Oslo, Oslo, Norway

Author contributions: Conceptualization: M. Bouchon. Investigation: M. Bouchon, D. Marsan, A. Socquet, J. Jara. Validation: S. Guillot, F. Renard. Writing: M. Bouchon, S. Guillot, D. Marsan, A. Socquet, J. Jara, F. Renard.

Abstract We analyze at a broad spatial scale the slab seismicity during one of the longest and best recorded foreshock sequences of a subduction earthquake to date: the M8.1 2014 Iquique earthquake in Chile. We observe the synchronisation of this sequence with seismic events occurring in the deep slab (depth ~100 km). We show that the probability that this synchronisation is obtained by chance is infinitesimal (<10⁻⁵), indicating that it is the result of a physical process taking place in the subduction. A mechanically logical explanation for this synchronicity seems to be the presence of fluid connections between the intermediate-depth range of the slab and the shallow seismogenic zone where foreshocks occur. These connections could be in the form of transient fluid channels in which bursts of pressure pulses would propagate, or localized high permeability paths along the plate interface in which pore-pressure waves would travel. It suggests that, like for the 2011 Tohoku earthquake, the deep slab was involved in the nucleation process of the Iquique earthquake. These observations may seem surprising but they are in line with the short-lived pulse-like channelized water escape from the dehydration zone predicted by recent studies in slab mineralogy and geochemistry.

Non-technical summary In 2014 a large earthquake (M8.1) occurred in the North Chile subduction. This earthquake was preceded by an intense foreshock crisis which lasted for nine months. We analyze here this foreshock activity and we observe that it is synchronised with seismic activity occurring deep (~100 km) in the slab, far down-dip from the foreshock locations and below the future rupture zone of the earthquake. As this deep seismic activity is thought to be associated with the dehydration of slab minerals and the release of water, it suggests that rapid water ascent from the dehydration zone may have triggered the foreshocks. Other possible mechanisms for this synchronicity of foreshocks with activity deep in the slab are discussed.

1 Introduction

Although it is still a controversial subject, an increasing number of observations support that broad spatial interactions occur in slabs. The rapidity and the scale of some of the interactions reported (Bouchon, 2016, 2022; Panet et al., 2018, 2022; Bedford, 2020; Bouih et al., 2022; Karabulut et al., 2022; Rousset et al., 2023) challenge our present understanding of slab dynamics and raise questions about the mechanism of communication across long (100km or more) distances. We analyze here, at a broad spatial scale, the long and wellrecorded foreshock sequence which preceded the M8.1 2014 Iquique earthquake in the North Chile subduction. Signs of short-term and long-term correlations between shallow and deep seismic activities there have been previously reported (Bouchon, 2016; Jara et al., 2017). We present here more detailed observations expanded to the whole foreshock crisis which lasted for nine months. The earthquake broke the Nazca/Southidentified as a major seismic gap (Madariaga, 1998). Although its foreshock sequence has attracted considerable attention Ruiz (2014); Schurr (2014); Kato and Nakagawa (2014); Kato et al. (2016); Lay et al. (2014); Bedford et al. (2015); Meng et al. (2015); Duputel (2015), these investigations concerned activity in the the seismogenic zone, which in subducting plates is limited to ~50 km depth. Below, the plate boundary slips almost continuously due to ductile deformation at elevated temperature. Megathrust earthquakes break the seismogenic zone but are not thought to extend much deeper. Below ~60 km, another type of seismic event, however, occurs in the descending plate. These events, termed intermediate-depth earthquakes, take place not along the interface but inside the cold core of the slab. They are believed to be linked to the metamorphic dehydration of slab minerals.

American plate interface in a region which had been

*Corresponding author: michel.bouchon@univ-grenoble-alpes.fr

Handling Editor: Christie Rowe Copy & Layout Editor: Kirsty Bayliss Signed reviewer(s):

Production Editor:

Gareth Funning

Pascal Audet and Cailey Condit

> Received: May 3, 2023 Accepted: September 5, 2023 Published: October 10, 2023

2 Synchronicity of Foreshocks with Activity Down-Dip in Slab

The pre-Iquique activity begins to be noticeable ~9 months before the mainshock (Kato et al., 2016; Socquet, 2017; Jara et al., 2017). The first phase of foreshock activity occurs in July-August 2013. It begins north of the epicenter and spreads southwards up to ~80 km from it in following weeks (Fig. 1). A quiescence period follows for a few months (September-December 2013, Aden-Antoniow et al., 2020). A second phase of foreshocks starts ~120 km south from the epicenter in January 2014, three months before the earthquake, and in the following weeks a broad slab segment is activated (Fig. 1). This activity intensifies on March 16, two weeks before the earthquake, when a M6.7 shock occurs. The evolution of this activity has been interpreted as migrating slow slip which is supported by GPS and tilt observations (Socquet, 2017; Boudin, 2021). The spatial extent of the foreshock zone, about 180 km, is intriguing as well as the rapidity with which seismicity spreads over a plate interface known to have been locked for decades (Madariaga, 1998; Chlieh, 2011; Metois et al., 2016).

We first study the relatively large (M > 4) events occurring in the subduction before the earthquake. We use for this the national catalog made by the Centro Seismologico Nacional of Chile (CSN, www.sismologia.cl, www.isc.ac.uk), whose completeness magnitude is around M4 (Jara et al., 2017). Whenever available (generally around and above magnitude 5), we use for these events the moment magnitudes published by the Global Centroid Moment Tensor (GCMT) Project (www.globalcmt.org, Ekström et al., 2012), which is the reference for large earthquakes worldwide.

In Fig. 2 we present the timing and magnitude of the shallow (depth<40 km) and deep (80 km<depth<125 km) earthquakes within increasing radial distance range (160 km, 170 km, and 200 km) from the future epicenter. The large depth separation differentiates shallow events (i.e. foreshocks) associated with the slip of the slab and intermediate-depth events associated with its internal deformation. The lower depth limit (125 km) has little effect as few events are deeper during the periods considered. Beyond this limit, events are far from stations and catalog resolution degrades. The first epicentral distance range considered is 160 km (Fig. 2a). At lower radii from the mainshock epicenter, the deep slab is not yet sampled because of the low dip of the slab. Within this distance range, the deep slab volume sampled lies directly down-dip from the epicentral zone. Fig. 2a shows that deep activity there is confined to the two foreshock crises and that one deep event shortly precedes (1 day) the intensification of the foreshock crisis which will lead to the earthquake two weeks later. Fig. 2b extends the exploration range to 170 km and focuses on the period preceding and including the first crisis. At this range where a larger volume of deep slab is sampled, a correlation emerges between deep and shallow activities. Fig. 2c extends the exploration range to 200 km and focuses on the period around the second crisis and on the largest shallow (M>4) and deep (M>4.5) events, the higher magnitude cut-off used for

2

deep events reflecting the higher level of deep background activity present in this subduction. At this distance range a time correlation between the two activities emerges clearly. The wider exploration needed for a correlation to emerge during the second crisis seems consistent with the broader spatial extent of this crisis (Fig. 1).

In statistics the two time series displayed in Fig. 2, are termed temporal point processes. To estimate the probability that one temporal point process (A) is dependent on the other one (B), a distribution of interevent times is constructed by fixing the events from series (B) and measuring the time from each event in (A) to the closest event in (B). This method is described in (Galbraith et al., 2020). Probability is calculated by fixing the times of the deep events, drawing randomly the times of the shallow events, and comparing their mean interevent time with the one observed. In doing so we do not make any hypothesis on any of the properties of the two time series. We simply look if the interevent time observed is due to random chance or if it is an intrinsic property of the data. The application of the method to seismic sequences is straightforward and described in Bouchon (2022). In Fig. 2b (first crisis) the chance probability that shallow events (i.e. foreshocks) are as closely synchronized with the occurrence of deep events is $< 10^{-5}$ (more than 100,000 random draws of the 9 M>4 shallow events are required to reach an interevent time with the 7 deep events present as small as the one observed). A similarly small chance probability $< 10^{-5}$ that shallow events occurring during the second crisis (Fig. 2c) would be as closely synchronized with deep events located within 200 km of epicentral distance is obtained. The combined probability that shallow events would be as closely synchronized with deep events below during the two foreshock crises is thus infinitesimal. The smallness of the values may seem surprising but it likely reflects the burst-like characteristic of the seismicity: As shown in Fig. 3, a burst is not simply made up of one shallow and one deep event, but usually of a multiplicity of them interweaved together within a short time, a characteristic difficult to be reproduced by a random process.

Using the catalog of Sippl et al. (2018a) for North Chile, which decreases the magnitude of completeness to ~2.7, we can explore shallow and deep activities at lower magnitude. Because of the high level of deep background activity below magnitude 4 in the subduction and the long duration and broad spatial extent of the crisis, we focus on the period when the first foreshock activity is the most intense and on the subduction segment where this activity takes place (Fig. 3). To interpret this figure, one has to realize that deep activity is continuously present in this zone, regardless of the occurrence or not of foreshocks. Consequently, if some interaction occurs between deep and shallow activities it probably does not involve all the deep population. Furthermore each family of events has necessarily dynamics of its own and smaller events may be aftershocks of the larger ones. A notable feature of Fig. 3 is that shallow activity is usually accompanied by deep activity. Calculating, as in Fig. 2, the chance probability



Figure 1 Location of the large (M>4) foreshocks showing their broad north-south spatial extent along the strike of the slab. All events with depth <40 km in the CSN catalog in the year preceding the earthquake up to the large M6.7 foreshock are shown. Most activity occurs in two crises. Circle size increases with magnitude. Star is the epicenter. Arrow indicates plate convergence direction (Vigny et al., 2009). Contour lines show slab interface depth (Hayes et al., 2012). Dark blue color marks the trench. White line shows the subduction segment considered in Fig. 3.

that the interevent time between the shallow (26 events) and the deep (22 events) occurrences is as small as the one observed yields a value $< 10^{-5}$.

An intriguing feature of Figs.2 and 3 is the burst-like occurrence of the events: The largest deep and shallow events occur in packets of short duration and, as can be seen in Fig. 3 (e.g. at -282, -266, -252 days etc), multiple deep and shallow events are often interweaved together within a burst. This complexity prevents the reading of a simplistic chronology between deep and shallow events.

3 Seismic Links indicative of Water Channels?

Fig. 4 shows where the M>4 shallow and deep events which make up the eight largest bursts (Fig. 2b-c) occur. It shows that during each burst, the deep events tend to occur nearly down-dip from the shallow events. This suggests a move along slab dip of the source of slip/deformation during each burst. The rapidity of this along dip move would be comparable to the migrating speed of tremors (Shelly et al., 2007; Ide, 2010; Ghosh, 2010; Gomberg, 2010; Peng and Gomberg, 2010; Beroza and Ide, 2011). The jumps of activity observed along subduction strike from one burst to the next are also tremor characteristics (Kao et al., 2007; Shelly et al., 2007; Ghosh, 2010).

Fig. 4 also displays the slab seismicity down to small magnitude during the entire 9-month long foreshock crisis. This seismicity map is made with the Sippl et al. (2018a) catalog. As we are interested in imaging possible seismic connections, it emphasizes clustered seismicity relative to isolated events. The spatial distribution of seismic events at depths of ~20-80 km, between the shallow and deep earthquake zones, is apparently aligned in lineaments parallel and oblique to the subduction interface dip direction. It supports the presence of seismic links connecting the shallow M>4 events to the deep slab. It also shows the tendency of these links to converge spatially towards the foreshock clusters and the epicenter. The paths these links define are



Figure 2 (a) Timings of all M>4 shallow and deep events located within 160 km from the epicenter in the year leading to the earthquake from the CSN catalog. The last event is the mainshock. (b) Increasing the epicentral distance to 170 km and focusing on the period before and during the first crisis. (c) Increasing the epicentral distance to 200 km and focusing on the largest shallow and deep events of the second crisis. Shallow events after the M6.7 foreshock are not shown because they are dominated by its own aftershocks. Periods of activity are indicated.

complex, sometimes multiple, but their long range continuity is notable.

Another illustration of seismic links between deep and shallow slab activities is presented in Fig. 5. This figure is made with the catalog of Aden-Antoniow et al. (2020) which uses a similar set of stations in North Chile as the Sippl et al. (2018a) catalog and has a comparable magnitude of completeness. It displays the seismicity pattern in August 2013 - the period when the first foreshock crisis is particularly intense. It shows the presence during this period of two seismic links connecting the future epicenter and the strongest foreshock cluster (M>5) to the locations of the largest intermediate-depth earthquakes (M>5) of this crisis.

4 Discussion

The presence of large volumes of water in subduction zones has long been documented (Raleigh and Paterson, 1965; Peacock, 1990; Green and Houston, 1995; Kirby et al., 1996; Hacker et al., 2003; Kawakatsu and Watada, 2007; Rondenay et al., 2008; Kawano et al., 2011; van Keken et al., 2011; John, 2012; Abers et al., 2013; Angiboust et al., 2014; Guillot et al., 2015; Plümper et al., 2017; Sippl et al., 2018b; Shapiro et al., 2018; Contreras-Reyes, 2021). The link between high-pressured fluids and seismic activity has itself long been recognized (Sibson, 1992; Miller et al., 1996).

The deep events in each burst occur in the depth range of 70 to 120 km where antigorite serpentine breaks down releasing the largest amount of water. Once released, water escape from the deep slab is thought to occur through transient channels (Miller et al., 2003; John, 2012; Angiboust et al., 2014; Plümper et al., 2017; Taetz et al., 2018).

The present observations are consistent with a mechanism involving the translation of pressure pulses in fluid-filled channels. The burst-like nature of the seismic activity would indicate that pressure propagation and fluid flow are very intermittent. This transient characteristic seems mechanically logical, with channels opening during overpressure passage and closing

SEISMICA | RESEARCH ARTICLE | Observation of a Synchronicity between Shallow and Deep Seismic Activities during the Foreshock Crisis Preceding the Iquique Megathrust Earthquake



Figure 3 Timings of shallow and deep events in the subduction segment where the most intense activity of the first foreshock crisis takes place. This segment (white line in Fig. 1) extends from 40 km south to 80 km south of the epicenter and its limits are aligned with the plate convergence direction. All the shallow (depth<40 km) and deep (80 km<depth<125 km) events from Sippl et al. (2018a) catalog occurring on this segment during the period considered are presented. The segment and period investigated correspond to the occurrence of the second and third bursts in Fig. 2b. Slight differences in magnitude relative to Fig. 2 are catalog differences. What is notable is that shallow activity is closely synchronized with some deep activity.

as soon as fluid pressure locally in the channel drops below local confining pressure. The along-dip organization of the bursts denotes an along-dip orientation of the channels, which probably reflects the strong downslip corrugation of the Nazca slab interface (Soto, 2019). Such corrugations have been recently proposed to act as fluid conducts (Edwards, 2018). The occurrence of the events in packets of short duration, including both shallow and deep events, often interweaved together, suggests that they are associated with the updip and downdip propagation of pressure pulses. While surges of overpressured fluids in the seismogenic zone are probably producing the foreshocks, they are accompanied by decompression pulses propagating downdip.

Another clear characteristic of the seismic activity is its long remarkable extension along the strike of the subduction (Figs. 1, 4). This long extension of the activity does not evolve in a continuous fashion but occurs in jumps. For instance, after ~4 months of quiescence, the second crisis begins suddenly in early January ~150 km away from where the first crisis had started and 50 km beyond the zone where foreshocks had previously occurred. The activity was strong there for a few days, then completely disappeared and by the end of January, foreshock activity had jumped back to a zone close to where it initiated.

The major characteristics that are observed, the rapidity of the up-dip/down-dip interactions, the jumps of the activities along subduction strike, the broad width of the subduction zone involved are not characteristics unseen before. These same characteristics have long been reported for tremors (e.g. Shelly et al., 2007; Kao et al., 2007; Ide, 2010; Ghosh, 2010; Gomberg, 2010; Peng and Gomberg, 2010; Beroza and Ide, 2011). What is novel, here and before the Tohoku earthquake, are the very long range and the depth reach of these phenomena as well as the relatively large magnitude of the seismic events produced.

One may question the existence of physical fluid channels at the depths considered. Their presence in the dehydration zone itself, however, is observed in exhumed rocks originating from this zone and is now well



Figure 4 Location of M>4 shallow and deep events (large colored symbols) occurring during the eight largest foreshock bursts. Each color/symbol represents one burst (dates for each burst indicated at top of map). Last burst ends with the lquique earthquake (large star). Shallow events are below sea, deep events below land. Shallow and deep events in each burst occur nearly along slab dip from each other. Superposed on the map is all the seismic activity during the entire crisis (July 2013 – March 2014) obtained from Sippl et al. (2018a) catalog (black dots: depth <40 km, blue dots: depth>40 km). The most spatially-clustered events (events with at least 3 neighbors within 10 km distance) are the larger dots.

documented (John, 2012; Angiboust et al., 2014; Plümper et al., 2017; Taetz et al., 2018) but direct observation on how these fluids migrate afterwards is lacking. Fig. 4 shows the presence of near continuous seismic paths connecting the foreshock zones to the locations of the largest intermediate depth events. The significance of these paths may at first be doubted on the ground that they are complex and multiple, but their convergence towards the foreshock and epicenter locations is clear and at least intriguing. The significance of the snapshot image of Fig. 5 might be also doubted because its statistical significance is difficult to assess, but it shows two clear seismic paths between the shallow and deep activities during one of the most active months of the foreshock crisis. The propagation of pore-pressure waves or porosity waves along or near the plate interface may be an alternative to the strong spatial localization of fluid flow of a channel model. Cruz-Atienza et al. (2018) have shown theoretically that tremor migration and speed can be explained by the propagation along the plate interface of non-linear pore-pressure waves under conditions that the interface is treated as a damage shear



Figure 5 Seismic activity (black dots) in May (left) and August (right) 2013 from Aden-Antoniow et al. (2020) catalog. Only the events occurring near the slab interface and below are shown. Size of the dots increases with magnitude. Superposed on the maps are the locations of the largest (M>5) foreshocks (red dots) and intermediate-depth earthquakes (blue dots) during the first foreshock crisis (July-August 2013). Seismic paths, not present or recognizable in May, before the crisis begins, link in August the most intense intermediate-depth earthquake zone of this crisis (three blue dots) to the strongest foreshock cluster (two red dots) and to the future epicenter (red star).

zone with strong permeability anisotropy. The seismic paths observed here could then be following the zones of highest permeability/highest shear deformation at or near the plate interface.

If one accepts that fluid/pressure circulation is the motor of the slab seismic activity observed during the foreshock crisis, one intriguing question is why, in such a short time (a few months), overpressured pulses/fluids would ascend from different distant places spanning such a long segment of the subduction. One possible mechanism would be the existence of connections between the deep rock reservoirs where water from dehydration is thought to be stored, so that pressure changes in one would affect others. Another possible mechanism could be a rapid deformation or slip of the slab, too small or too deep to be detected geodetically, but of broad spatial extent, which could disturb the slab interface and the fluid present at depth. This probably would imply that the whole slab interface is nearing threshold stress so that effective stress limit is reached nearly simultaneously along strike for ~200 km. One may also wonder if the foreshock crisis could be driven by the updip pressures from slow slip events occurring at depth.

The present study is limited to the 9 months-long duration of the Iquique foreshock period, but it may be of interest that one of the largest intermediate-depth earthquake in instrumental time in Chile, the 2005 M7.8 Tarapaca earthquake, occurred 9 years before, precisely down-dip below the area which was to rupture during the Iquique earthquake (Jara, 2018; Ruiz and Madariaga, 2018). Although seismic instrumentation in North Chile at the time was too sparse to conduct the same study as the one done here, it is notable that this earthquake produced a long-term decrease in GPS eastward velocities in the region, interpreted as a decrease in plate coupling (Jara et al., 2017). This situation appears surprisingly comparable to the occurrence of the 2003 M7.1 intermediate-depth earthquake down-dip below the Tohoku epicentral zone 8 years earlier. Although the mechanisms by which slab dehydration, and its accompanying water release, induce intermediate-depth earthquakes are still debated (e.g. van Keken et al., 2012; Abers et al., 2013; Prieto, 2013; Poli and Prieto, 2014; Ferrand, 2017; Gasc et al., 2017; Cabrera et al., 2021), the association of the two seems now well established.

The interpretation of our observations seems supported by recent studies in subduction zone mineralogy and geochemistry which predict the short-lived pulse-like channelized water escape from the dehydration zone (John, 2012; Angiboust et al., 2014; Plümper et al., 2017; Taetz et al., 2018).

5 Conclusion

The present observations show the synchronisation of the foreshock activity which preceded the Iquique megathrust earthquake in Chile with seismic activity occurring below the foreshock locations in the intermediate-depth range of the slab. They also show the presence of near-continuous seismic links connecting the two activities. These characteristics are similar to the ones observed before the Tohoku earthquake, supporting that the same physical processes led to the two megathrust ruptures. The most logical interpretation of these observations in today's knowledge seems to be the rapid ascent of water from the slab dehydra-

tion zone.

Acknowledgements

We thank Christie Rowe for her extremely valuable comments and for her dedicated work in editing the manuscript. We also thank the two reviewers for their time and their numerous and helpful comments. We thank C. Sippl, B. Schurr, G. Asch, J. Kummerow, and F. Aden-Antoniow, C. Satriano, P. Bernard, N. Poiata, E. Aissaoui, J.P. Vilotte, W. Frank for making their catalogs publicly available. We thank the scientists, technicians and staff of the Centro Seismologico Nacional of Chile, the International Seismological Centre, and the Global Centroid Moment Tensor Project whose work made this study possible. This work benefited from discussions with C. Voisin, F. Brunet, M. Chlieh, M. Radiguet, P. Poli, E. Pathier, A. Walpersdorf.

Data and code availability

The data used in this study are open and available at: Sippl et al. (2018b), Aden-Antoniow et al. (2020), www.sismologia.cl, www.isc.ac.uk and www.globalcmt.org.

Competing interests

The authors declare that they have no competing interests.

References

- Abers, G., Nakajima, J., Keken, P., Kita, S., and Hacker, B. Thermalpetrological controls on the location of earthquakes within subducting plates. *Earth Plan. Sc. Lett*, 369:178–187, 2013. doi: https://doi.org/10.1016/j.epsl.2013.03.022.
- Aden-Antoniow, F., Satriano, C., Bernard, P., Poiata, N., Aissaoui, E., Vilotte, J., and Frank, W. Statistical analysis of the preparatory phase of the M8.1 lquique earthquake, Chile. J. Geophys. Res, 2019JB019337, 2020. doi: https://doi.org/10.1029/2019JB019337.
- Angiboust, S., Pettke, T., Hoog, J., Caron, B., and Oncken, O. Channelized fluid flow and eclogite-facies metasomatism along the subduction shear zone. *J. Petrol*, 55:883–916, 2014. doi: https://doi.org/10.1093/petrology/egu010.
- Bedford, J. Months-long thousand-kilometre-scale wobbling before great subduction earthquakes. *Nature*, 580:628–635, 2020. doi: https://doi.org/10.1038/s41586-020-2212-1.
- Bedford, J., Moreno, M., Schurr, B., Bartsch, M., and Oncken, O. Investigating the final seismic swarm before the lquique-Pisagua 2014 Mw 8.1 by comparison of continuous GPS and seismic foreshock data. *Geophys. Res. Lett*, 42:3820–3828, 2015. doi: https://doi.org/10.1002/2015GL063953.
- Beroza, G. and Ide, S. Slow earthquakes and nonvolcanic tremors. Annu. Rev. Earth Planet. Sci, 39:271–296, 2011. doi: https://doi.org/10.1146/annurev-earth-040809-152531.
- Bouchon, M. Potential slab deformation and plunge prior to the Tohoku, Iquique and Maule earthquakes. *Nature Geosci*, 9: 380–383, 2016. doi: https://doi.org/10.1038/ngeo2701.

- Bouchon, M. Observation of rapid long-range seismic bursts in the Japan Trench subduction leading to the nucleation of the Tohoku earthquake. *Earth Plan. Sc. Lett*, 594:117696, 2022. doi: https://doi.org/10.1016/j.epsl.2022.117696.
- Boudin, F. Slow slip events precursory to the 2014 Iquique earthquake, revisited with long-base tilt and GPS records. *Geophys. J. Int*, 228:2092–2121, 2021. doi: https://doi.org/10.1093/gji/ggab425.
- Bouih, M., Panet, I., Remy, D., Longuevergne, L., and Bonvalot, S. Deep mass redistribution prior to the 2010 Mw 8.8 Maule (Chile) Earthquake revealed by GRACE satellite gravity. *Earth Plan. Sc. Lett*, 584:117465, 2022. doi: https://doi.org/10.1016/j.epsl.2022.117465.
- Cabrera, L., Ruiz, S., Poli, P., Contreras-Reyes, E., Osses, A., and Mancini, R. Northern Chile intermediate-depth earthquakes controlled by plate hydration. *Geophys. J. Int*, 226:78–90, 2021. doi: https://doi.org/10.1093/gji/ggaa565.
- Chlieh, M. Interseismic coupling and seismic potential along the Central Andes subduction zone. *J. Geophys. Res*, 116(B12):1–21, 2011. doi: https://doi.org/10.1029/2010JB008166.
- Contreras-Reyes, E. Subduction zone fluids and arc magmas conducted by lithospheric deformed regions beneath the central Andes. *Scientific Reports*, 11(23078), 2021. doi: https://doi.org/10.1038/s41598-021-02430-9.
- Cruz-Atienza, V., Villafuerte, C., and Bhat, H. Rapid tremor migration and pore-pressure waves in subduction. *Nature Comm*, 2018. doi: 10.1038/s41467-018-05150-3.
- Duputel, Z. The Iquique earthquake sequence of April 2014: Bayesian modeling accounting for prediction uncertainty. *Geophys. Res. Lett*, 42:7949–7957, 2015. doi: https://doi.org/10.1002/2015GL065402.
- Edwards, J. Corrugated megathrust revealed offshore from Costa Rica. *Nature Geosci*, 11:197–202, 2018. doi: https://doi.org/10.1038/s41561-018-0061-4.
- Ekström, G., Nettles, M., and Dziewonski, A. The global CMT project 2004-2010: Centroid-moment tensors for 13,017 earthquakes. *Phys.Earth Planet. Inter*, page 200–201, 2012. doi: 10.1016/j.pepi.2012.04.002.
- Ferrand, T. Dehydration-driven stress transfer triggers intermediate-depth earthquakes. *Nature Comm*, 8:15247, 2017. doi: https://doi.org/10.1038/ncomms15247.
- Galbraith, C., Smyth, P., and Stern, H. Quantifying the association between discrete event time series with applications to digital forensics. *J. Roy. Stat. Soc*, Series A:0964–1998 20 183000, 2020. doi: http://doi.org/10.1111/rssa.12549.
- Gasc, J., Hilairet, N., Yu, T., Ferrand, T., Schubnel, A., and Y, W. Faulting of natural serpentine: Implications for intermediate-depth seismicity. *Earth Plan. Sc. Lett*, 474:138–147, 2017. doi: https://doi.org/10.1016/j.epsl.2017.06.016.
- Ghosh, A. Rapid continuous streaking of tremor in Cascadia. Geoch Geophys. Geosyst, 11, Q12010, 2010. doi: https://doi.org/10.1029/2010GC003305.
- Gomberg, J. Slow slip phenomena in Cascadia from 2007 and beyond: A review. *Geol. Soc. Am. Bull*, 122:963–978, 2010. doi: https://doi.org/10.1130/B30287.1.
- Green, H. and Houston, H. The mechanics of deep earthquakes. Ann. Rev. Earth Planet. Sci, 23:169–213, 1995. doi: https://doi.org/10.1146/annurev.ea.23.050195.001125.
- Guillot, S., Schwartz, S., Agard, P., Renard, B., and Prigent, C. Tectonic significance of serpentinites. *Tectonophysics*, page 1–19, 2015. doi: https://doi.org/10.1016/j.tecto.2015.01.020.
- Hacker, B., Peacock, S., Abers, G., and Holloway, S. Subduction factory 2. Are intermediate-depth earthquakes in subducting slabs

linked to metamorphic dehydration reactions? *J. Geophys. Res*, 108, 2003. doi: https://doi.org/10.1029/2001JB001129.

- Hayes, G., Wald, D., and Johnson, R. Slab1.0: A three-dimensional model of global subduction zone geometries. *J. Geophys. Res*, 117, B01302, 2012. doi: https://doi.org/10.1029/2011JB008524.
- Ide, S. Striations, duration, migration, and tidal response in deep tremor. *Nature*, 466:356–359, 2010. doi: https://doi.org/10.1038/nature09251.
- Jara, J. Kinematic study of lquique 2014 Mw8.1 earthquake; Understanding the segmentation of the seismogenic zone. *Earth Plan. Sc. Lett*, 503:131–143, 2018. doi: https://doi.org/10.1016/j.epsl.2018.09.025.
- Jara, J., Soquet, A., Marsan, D., and Bouchon, M. Long-term interactions between intermediate depth and shallow seismicity in North Chile subduction Zone. *Geophys. Res. Lett*, 44, 2017. doi: https://doi.org/10.1002/2017GL075029.
- John, T. Volcanic arcs fed by rapid pulsed fluid flow through subducting slabs. *Nature Geosci*, 5:489–492, 2012. doi: https://doi.org/10.1038/ngeo1482.
- Kao, H., Shan, S., Rogers, G., and Dragert, H. Migration characteristics of seismic tremors in the northern Cascadia margin. *Geophys. Res. Lett*, 34, L03304, 2007. doi: https://doi.org/10.1029/2006GL028430.
- Karabulut, H., Bouchon, M., and Schmittbuhl, J. Synchronization of small scale seismic clusters reveals large scale plate deformation. *Earth Planets Space*, 74(158), 2022. doi: https://doi.org/10.1186/s40623-022-01725-z.
- Kato, A. and Nakagawa, S. Multiple slow-slip events during a foreshock sequence of the 2014 Iquique, Chile Mw 8.1 earthquake. *Geophys. Res. Lett*, 41:5420–5427, 2014. doi: https://doi.org/10.1002/2014GL061138.
- Kato, A., Fukuda, J., Kumazawa, T., and Nakagawa, S. Accelerated nucleation of the 2014 Iquique, Chile Mw 8.2 earthquake. *Nature Sci. Rep*, 6:24792, 2016. doi: https://doi.org/10.1038/srep24792.
- Kawakatsu, H. and Watada, S. Seismic Evidence for Deep-Water Transportation in the Mantle. *Science*, 316:1468–1471, 2007. doi: https://doi.org/10.1126/science.1140855.
- Kawano, S., Katayama, I., and Okazaki, K. Permeability anisotropy of serpentine and fluid pathways in a subduction zone. *Geology*, 39:939–942, 2011. doi: https://doi.org/10.1130/G32173.1.
- Kirby, S., Engdahl, R., and Denlinger, R. Intermediate-depth intraslab earthquakes and arc volcanism as physical expressions of crustal and uppermost mantle metamorphism in subducting slabs. Subduction Top to Bottom, Am. Geophys. Un. Geophys. Monogr, 96:195–214, 1996. doi: https://doi.org/10.1029/GM096p0195.
- Lay, T., Yue, H., Brodsky, E., and An, C. The 1 April 2014 Iquique, Chile, Mw 8.1 earthquake rupture sequence. *Geophys. Res. Lett*, 41:3818–3825, 2014. doi: https://doi.org/10.1002/2014GL060238.
- Madariaga, R. Sismicidad de Chile. *Fisica de la Tierra*, 10:221–258, 1998.
- Meng, L., Huang, H., Bürgmann, R., Ampuero, J., and Strader, A. Dual megathrust behaviors of the 2014 Iquique earthquake sequence. *Earth Planet. Sci. Lett*, 411:177–187, 2015. doi: https://doi.org/10.1016/j.epsl.2014.11.041.
- Metois, M., Vigny, C., and Socquet, A. Interseismic coupling, megathrust earthquakes and seismic swarms along the Chilean subduction zone (38°-18°S. *Pure Appl. Geophys*, 173:1431–1449, 2016. doi: https://doi.org/10.1007/s00024-016-1280-5.
- Miller, S., Nur, A., and Olgaard, D. Earthquakes as a coupled shear stress-high pore pressure dynamical system. *Geophys. Res. Lett*, 23:197–200, 1996. doi: https://doi.org/10.1029/95GL03178.

- Miller, S., Zee, W., Olgaard, D., and Connolly, J. A fluid-pressure feedback model of dehydration reactions: experiments, modelling, and application to subduction zones. *Tectonophysics*, 370:241–251, 2003. doi: https://doi.org/10.1016/S0040-1951(03)00189-6.
- Panet, I., Bonvalot, S., Narteau, C., Remy, D., and Lemoine, J. Migrating pattern of deformation prior to the Tohoku-Oki earthquake revealed by GRACE data. *Nature Geosci*, 11, 2018. doi: https://doi.org/10.1038/s41561-018-0099-3.
- Panet, I., Narteau, C., Lemoine, J., Bonvalot, S., and Remy, D. Detecting preseismic signals in GRACE gravity solutions : Application to the 2011 Mw 9.0 Tohoku earthquake. *J. Geophys. Res*, 127, 2022. doi: https://doi.org/10.1029/2022JB024542.
- Peacock, S. Fluid processes in subduction zones. Science, 248:329–337, 1990. doi: https://doi.org/10.1126/science.248.4953.329.
- Peng, Z. and Gomberg, J. An integrated perspective of the continuum between earthquakes and slow-slip phenomena. *Nature Geosci*, 3:599–607, 2010. doi: https://doi.org/10.1038/ngeo940.
- Plümper, O., John, T., Podladchikov, Y., Vrijmoed, J., and Scambelluri, M. Fluid escape from subduction zones controlled by channel-forming reactive porosity. *Nature Geosci*, 10:150–155, 2017. doi: https://doi.org/10.1038/ngeo2865.
- Poli, P. and Prieto, G. Global and along-strike variations of source duration and scaling for intermediate-depth and deep-focus earthquakes. *Geophys. Res. Lett*, 41:8315–8324, 2014. doi: https://doi.org/10.1002/2014GL061916.
- Prieto, G. Seismic evidence for thermal runaway during intermediate-depth earthquake rupture. *Geophys. Res. Lett*, 40: 6064–6068, 2013. doi: https://doi.org/10.1002/2013GL058109.
- Raleigh, C. and Paterson, M. Experimental deformation of serpentinite and its tectonic implications. J. Geophys. Res, 70:3965–3985, 1965. doi: https://doi.org/10.1029/JZ070i016p03965.
- Rondenay, S., Abers, G., and P.E, K. Seismic imaging of subduction zone metamorphism. *Geology*, 36:275–278, 2008. doi: https://doi.org/10.1130/G24112A.1.
- Rousset, B., Campillo, M., Shapiro, N., Walpersdorf, A., Titkov, N., and Chebrov, D. The 2013 slab-wide Kamchatka earthquake sequence. *Geophys. Res. Lett*, 50, 2023. doi: 10.1029/2022GL101856.
- Ruiz, S. Intense foreshocks and a slow slip event preceded the 2014 Iquique Mw 8.1 earthquake. *Science*, 345:1165–1169, 2014. doi: https://doi.org/10.1126/science.1256074.
- Ruiz, S. and Madariaga, R. Historical and recent large megathrust earthquakes in Chile. *Tectonophysics*, 733:37–56, 2018. doi: https://doi.org/10.1016/j.tecto.2018.01.015.
- Schurr, B. Gradual unlocking of plate boundary controlled initiation of the 2014 Iquique earthquake. *Nature*, 512:299–302, 2014. doi: https://doi.org/10.1038/nature13681.
- Shapiro, N., Campillo, M., Kaminski, E., Vilotte, J., and Jaupard, C. Low-frequency earthquakes and pore-pressure transients in subduction zones. *Geophys. Res. Lett*, 43:11083–11094, 2018. doi: https://doi.org/10.1029/2018GL079893.
- Shelly, D., Beroza, G., and Ide, S. Complex evolution of transient slip derived from precise tremor locations in western Shikoku, Japan. *Geochem. Geophys. Geosyst*, 8, Q10014, 2007. doi: https://doi.org/10.1029/2007GC001640.
- Sibson, R. Implications of fault-valve behavior for rupture nucleation and recurrence, Tectonophysics 18,1031–1042, 1992. doi: https://doi.org/10.1016/0040-1951(92)90065-E.
- Sippl, C., Schurr, B., Asch, G., and Kummerow, J. Catalogue of Earthquake Hypocenters for Northern Chile Compiled from IPOC

(plus auxiliary) seismic stations. GFZ Data Services, 2018a. doi: 10.5880/GFZ.4.1.2018.001.

- Sippl, C., Schurr, B., Asch, G., and Kummerow, K. Seismicity structure of the northern Chile forearc from >100,000 double-difference relocated hypocenters. J. Geophys. Res. Solid Earth, 123:4063–4087, 2018b. doi: https://doi.org/10.1002/2017JB015384.
- Socquet, A. An 8 month slow slip event triggers progressive nucleation of the 2014 Chile megathrust. *Geophys. Res. Lett*, 44, 2017. doi: https://doi.org/10.1002/2017GL073023.
- Soto, H. Probing the Northern Chile Megathrust with Seismicity: The 2014 M8.1 Iquique Earthquake Sequence. J. Geophys. Res, 124:12935–12954, 2019. doi: https://doi.org/10.1029/2019JB017794.
- Taetz, S., John, T., Bröcker, M., Spandler, C., and Stracke, A. Fast intraslab fluid-flow events linked to pulses of high pore fluid pressure at the subducted plate interface. *Earth Plan. Sc. Lett*, 482: 33–43, 2018. doi: https://doi.org/10.1016/j.epsl.2017.10.044.
- van Keken, P., Hacker, B., Syracuse, E., and Abers, G. Subduction factory: Depth-dependent flux of H2O from subducting slabs worldwide. *J. Geophys. Res*, 116, B01401, 2011. doi: https://doi.org/10.1029/2010JB007922.
- van Keken, P., Kita, S., and Nakajima, J. Thermal structure and intermediate-depth seismicity in the Tohoku-Hokkaido subduction zones. *Solid Earth*, 3:355–364, 2012. doi: https://doi.org/10.5194/se-3-355-2012.
- Vigny, C., Rudloff, A., Ruegg, J., Madariaga, R., Campos, J., and Alvarez, M. Upper plate deformation measured by GPS in the Coquimbo gap. *Chile, Phys. Earth planet. Inter*, 175:86–95, 2009. doi: https://doi.org/10.1016/j.pepi.2008.02.013.

The article Observation of a Synchronicity between Shallow and Deep Seismic Activities during the Foreshock Crisis Preceding the Iquique Megathrust Earthquake © 2023 by Michel Bouchon is licensed under CC BY 4.0.